

GOPEN ACCESS

Citation: Sié A, Ouattara M, Bountogo M, Dah C, Ouedraogo T, Boudo V, et al. (2024) Single-dose azithromycin for infant growth in Burkina Faso: Prespecified secondary anthropometric outcomes from a randomized controlled trial. PLoS Med 21(1): e1004345. https://doi.org/10.1371/journal. pmed.1004345

Academic Editor: Zulfiqar A. Bhutta, The Hospital for Sick Children, CANADA

Received: October 13, 2023

Accepted: January 10, 2024

Published: January 23, 2024

Copyright: © 2024 Sié et al. This is an open access article distributed under the terms of the <u>Creative</u> Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: De-identified source data for the study are available through the Open Science Framework (https://osf.io/y9wtr/). DOI 10. 17605/OSF.IO/Y9WTR.

Funding: This study was funded by the Bill and Melinda Gates Foundation (OPP1187628, PI: Lietman). The funders played no role in the study design, data collection, analysis, decision to publish, or preparation of the manuscript. **RESEARCH ARTICLE**

Single-dose azithromycin for infant growth in Burkina Faso: Prespecified secondary anthropometric outcomes from a randomized controlled trial

Ali Sié ^{1,2}*, Mamadou Ouattara¹, Mamadou Bountogo¹, Clarisse Dah¹, Thierry Ouedraogo¹, Valentin Boudo¹, Elodie Lebas², Huiyu Hu², Benjamin F. Arnold^{2,3,4}, Kieran S. O'Brien^{2,3,4,5}, Thomas M. Lietman^{2,3,4,5}, Catherine E. Oldenburg^{2,3,4,5}*

 Centre de Recherche en Santé de Nouna, Nouna, Burkina Faso, 2 Francis I Proctor Foundation, University of California, San Francisco, United States of America, 3 Department of Ophthalmology, University of California, San Francisco, United States of America, 4 Institute for Global Health Sciences, University of California, San Francisco, United States of America, 5 Department of Epidemiology & Biostatistics, University of California, San Francisco, United States of America

* sieali@yahoo.fr (AS); catherine.oldenburg@ucsf.edu (CEO)

Abstract

Background

Antibiotic use during early infancy has been linked to childhood obesity in high-income countries. We evaluated whether a single oral dose of azithromycin administered during infantwell visits led to changes in infant growth outcomes at 6 months of age in a setting with a high prevalence of undernutrition in rural Burkina Faso.

Methods and findings

Infants were enrolled from September 25, 2019, until October 22, 2022, in a randomized controlled trial designed to evaluate the efficacy of a single oral dose of azithromycin (20 mg/kg) compared to placebo when administered during well-child visits for prevention of infant mortality. The trial found no evidence of a difference in the primary endpoint. This paper presents prespecified secondary anthropometric endpoints including weight gain (g/ day), height change (mm/day), weight-for-age Z-score (WAZ), weight-for-length Z-score (WLZ), length-for-age Z-score (LAZ), and mid-upper arm circumference (MUAC). Infants were eligible for the trial if they were between 5 and 12 weeks of age, able to orally feed, and their families were planning to remain in the study area for the duration of the study. Anthropometric measurements were collected at enrollment (5 to 12 weeks of age) and 6 months of age. Among 32,877 infants enrolled in the trial, 27,298 (83%) were followed and had valid anthropometric measurements at 6 months of age. We found no evidence of a difference in weight gain (mean difference 0.03 g/day, 95% CI –0.05 to 0.06), WAZ (mean difference -0.004 SD, 95% CI –0.03 to 0.02), WLZ (mean difference 0.001 SD, 95% CI –0.03 to 0.03),

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: CI, confidence interval; LAZ, lengthfor-age Z-score; MUAC, mid-upper arm circumference; SD, standard deviation; WAZ, weight-for-age Z-score; WHO, World Health Organization; WLZ, weight-for-length Z-score. LAZ (mean difference -0.005 SD, 95% CI -0.03 to 0.02), or MUAC (mean difference 0.01 cm, 95% CI -0.01 to 0.04). The primary limitation of the trial was that measurements were only collected at enrollment and 6 months of age, precluding assessment of shorter-term or long-term changes in growth.

Conclusions

Single-dose azithromycin does not appear to affect weight and height outcomes when administered during early infancy.

Trial registration

ClinicalTrials.gov NCT03676764

Author summary

Why was this study done?

- Antibiotics have been linked to obesity in children in nonrandomized studies in highincome settings.
- In low-income settings, malnutrition is a strong risk factor for child mortality, and interventions that increase weight gain may be beneficial.
- Single-dose azithromycin has previously been shown to reduce child mortality in sub-Saharan Africa, and it is possible that some of this effect is via weight gain.

Why did the researchers do and find?

- We conducted a randomized controlled trial in which infants aged 5 to 12 weeks were randomly assigned to a single dose of azithromycin or a matching placebo.
- The primary outcome of the parent trial was all-cause mortality, and the trial found no evidence of a benefit of azithromycin for prevention of mortality.
- Infants were followed until 6 months of age and had anthropometric measurements, including height, weight, and mid-upper arm circumference, collected.
- We found no evidence of a difference in any anthropometric endpoint in infants randomized to azithromycin compared to placebo.

What do these findings mean?

- Single-dose azithromycin does not appear to affect growth when administered to healthy infants who are 5 to 12 weeks of age.
- Growth outcomes were only measured at 6 months of age, and thus, any shorter or longer-term effects of azithromycin would not be detected by this study.

• This study only considered a single dose of azithromycin, as is used for prevention of child mortality and trachoma, and thus, it is possible that higher doses or longer duration of antibiotics could have a different effects on child growth.

Introduction

Antibiotics are known to modulate the gut microbiome and are thought to be growth promoting in humans [1–5]. Some randomized controlled trials evaluating antibiotics in children with acute malnutrition, diarrhea, HIV, and other comorbidities have found increases in weight gain with children receiving antibiotics compared to placebo [6,7], while others have found no difference in growth outcomes [8]. In children who are HIV-exposed and uninfected, cotrimoxazole was associated with reduced risk of stunting [9]. Observational studies in high-income settings have suggested that early childhood antibiotic exposure is a risk factor for obesity and that there may be a dose-dependent response between antibiotic exposure and body mass index [10,11]. Most studies considering the effect of antibiotics on child growth have considered multiday antibiotic regimens, although any dose-dependent effect of antibiotics on growth is unclear.

In low- and middle-income countries, undernutrition contributes to approximately 45% of child deaths [12]. Underweight (low weight-for-age Z-score (WAZ)) is a strong predictor of mortality in children under 5 years of age and recently has been identified as one of the strongest anthropometric predictors of mortality among infants under 6 months of age [13–15]. Interventions to improve nutritional status in young infants must not interfere with exclusive breastfeeding, and thus, interventions such as supplemental or fortified foods are not appropriate in this age group. Mass drug administration with single-dose azithromycin has previously been shown to reduce all-cause childhood mortality among children 1 to 59 months of age in sub-Saharan Africa, with the largest effects in children aged 1 to 11 months in the highest mortality settings [16]. If antibiotics are growth promoting in the general population of infants, one potential mechanism of azithromycin for prevention of mortality may be via increased weight gain and reduced risk of underweight.

A previous study of single-dose azithromycin administered during the neonatal period for prevention of mortality found no overall effect of neonatal (age 8 to 27 days) azithromycin on 6-month mortality, but the study was underpowered due to a lower than anticipated mortality rate [17]. A single case of infantile hypertrophic pyloric stenosis was identified in that study in a male infant who received azithromycin [17]. In that trial, there was no evidence of an effect of single-dose azithromycin on infant anthropometric outcomes in the overall trial population [18]. These results were consistent with studies of growth among children under 5 years of age receiving single-dose azithromycin for trachoma control or prevention of mortality in cluster randomized trials [19–22]. In a subgroup analysis of the neonatal trial, neonates who were low birthweight (<2,500 g) and also underweight at enrollment (WAZ <-2 at age 8 to 27 days) had reduced odds of underweight at 6 months of age if they received azithromycin compared to placebo [23]. This trial excluded infants who were less than 2,500 g at enrollment (second to fourth weeks of life) for safety reasons, and thus, subgroups defined by the smallest infants were often underpowered. However, these results led to the hypothesis that single-dose azithromycin may not cause weight gain in the general population but may be growth promoting in subgroups of infants that are more vulnerable.

Here, we report data from a randomized controlled trial of over 30,000 infants in Burkina Faso of a single dose of azithromycin or placebo between the age of 5 and 12 weeks for prevention of infant mortality to evaluate effects of azithromycin on growth endpoints at 6 months of age. In the current trial, there were no enrollment exclusions based on weight, allowing for evaluation of a wider range of baseline anthropometric status. Based on the results of the neonatal trial [18], we hypothesized that overall, there would be no effect of azithromycin on growth endpoints but that there would be some benefit for infants with anthropometric deficits (e.g., underweight, stunting, and wasting) at enrollment.

Methods

Study setting

This study was conducted in the Nouna and Banfora districts of Burkina Faso. Burkina Faso is a landlocked country in West Africa in the Sahel that experiences seasonal rainfall from approximately July through October. Participants in the study were enrolled in Nouna District, in the northwest near the Mali border, and in Banfora, in the south.

Trial methods

Complete methods for the trial have been previously reported [24]. In brief, the study was a 1:1 randomized placebo-controlled trial evaluating the efficacy of a single dose of azithromycin (20 mg/kg) or matching placebo for prevention of mortality by 6 months of age. Infants were enrolled from September 22, 2019, until October 22, 2022. The trial's primary outcome, all-cause mortality, has been reported separately, as has adverse event data [25]. There was no evidence of a difference in the primary endpoint of the trial. This paper presents prespecified secondary endpoints, including weight gain (g/day), height change (mm/day), WAZ, weight-forlength Z-score (WLZ), and length-for-age Z-score (LAZ). The trial was reviewed and approved by the Institutional Review Board at the University of California, San Francisco, the Comite Institutionnel d'Ethique at the Centre de Reserche en Santé de Nouna, the Comite d'Ethique pour la Recherche en Santé in Ouagadougou, and the Comité Technique d'Examen des Demandes d'Autorisation d'Essais Cliniques in Ouagadougou. Written informed consent was obtained from the caregiver of each participant. The study was conducted in accordance with the Declaration of Helsinki.

The parent trial was designed to establish the efficacy of single-dose azithromycin administered during routine well-child visits (e.g., vaccination visits) during early infancy for prevention of all-cause mortality at 6 months of age. Infants were eligible for enrollment if they were between 5 and 12 weeks of age, were able to feed orally, their families were planning to stay in the study area for the full 6-month period of the study, and with appropriate caregiver consent. Infants were recruited directly in their community of residence via vaccine outreach teams that visit each community monthly, or in clinics during well-child visits [26]. During community outreach, mobilizers visited each study community to inform families that the study team if interested in participating. In clinics, families were informed of the study and were directed to the study team if interested in participation.

Interventions

Participants were randomized in a 1:1 fashion without stratification or blocking to a single oral dose of azithromycin (20 mg/kg) or an equivalent volume of matching placebo. The placebo was identical to the azithromycin in taste, smell, and appearance. Study medications were

donated by Pfizer (Pfizer, New York, NY). Participants were randomized and received their study medication after all baseline assessments, including anthropometric measurements, were completed. All study medication doses were directly observed.

Anthropometric measurements

Anthropometric measurements were collected at baseline and at 6 months of age. Infants were weighed using a digital infant scale (ADE M112600U scale, Hamburg, Germany). Caregivers were instructed to remove bulky items of clothing prior to weighing each infant. The scale was standardized and calibrated each morning using a known test weight. Length was measured using a ShorrBoard (Weigh and Measure, Olney, MD). Due to the potential for measurement error, length measurements were taken in triplicate and the median was used for analysis. Mid-upper arm circumference (MUAC) was measured using a standard MUAC tape. WAZ (a marker of underweight), WLZ (a marker of wasting), and LAZ (a marker of stunting) were calculated using 2006 World Health Organization (WHO) reference standards [27]. Underweight was defined as WAZ <-2, wasting as WLZ <-2, and stunting as LAZ <-2. At 6 months of age, we additionally considered MUAC < 12.5 cm to be a sign of wasting.

Outcomes

All prespecified secondary anthropometric endpoints at 6 months of age are included in this report, including weight gain (in g/day), height change (in mm/day), weight, length, WAZ, WLZ, LAZ, and MUAC. We additionally evaluated dichotomized post hoc outcomes of underweight, wasting defined by WLZ, wasting defined by MUAC, and stunting. Per our prespecified analysis plan, infants with values outside of WHO child growth standards for WAZ (-6 to +5 SD), LAZ (-6 to +6 SD), or WLZ (-5 to +5 SD) were excluded from the analysis. Additional endpoints included mortality, serious adverse events, hospitalizations, and clinic visits and are reported separately [25].

Subgroup analyses

A series of non-prespecified subgroup analyses were conducted by the child's age in months, sex, season of enrollment, at baseline underweight (WAZ < -2), wasting (WLZ < -2), and stunting (LAZ < -2) was assessed for all outcomes.

Statistical methods

We used an intention-to-treat analysis for all outcomes. The prespecified analysis plan included superiority analyses for all outcomes. The randomization was not stratified, and the prespecified analysis plan did not include any adjustments for clustering. For weight gain (g/ day) and height change (mm/day), we used an unadjusted linear regression model with a term for the infant's randomized treatment assignment. For weight, length, WAZ, WLZ, LAZ, and MUAC, we used a linear regression model with a term for the randomized treatment assignment and adjusted for the baseline measure of each outcome variable. For underweight, wasting, and stunting outcomes, we used a logistic regression model with a single term for the infant's randomized treatment assignment. For each subgroup for each outcome, a similar model was used as in the main analysis, with an interaction term between the subgroup and randomized treatment assignment to assess for the presence of effect modification on the additive scale. P < 0.05 was considered statistically significant, and all tests were two-sided. All analyses were run in R version 4.3.1 (The R Foundation for Statistical Computing, Vienna Austria).

Results

Of 32,877 infants enrolled in the trial, 16,416 were randomized to azithromycin and 16,461 to placebo (**Fig 1**). Of these, 13,641 (83%) in the azithromycin arm and 13,657 (83%) in the placebo arm were included in the analysis. Reasons for being excluded from the analysis were that the infant had no anthropometric measurement (72%), the infant was lost to follow-up (13%), the infant was measured out of the prespecified visit window (8%), or the anthropometric measurement was not valid per WHO Child Growth Standards (7%; **Fig 1**). At baseline, infants were a median of 49 days old and 49% were female (**Table 1**). Baseline anthropometric measurements were similar between the 2 arms; median WLZ was -0.2 (standard deviation (SD), 1.5), median WAZ was -0.6 (SD 1.2), and median LAZ was -0.5 (SD 1.3) in each arm (**Table 1**). Baseline characteristics of infants who were and were not included in the analysis were similar (**Table A in S1 Appendix**).

At 6 months, we found no evidence of a difference in any anthropometric endpoint between the azithromycin and placebo arms in the overall population (Table 2). The mean value for each continuous anthropometric endpoint was nearly identical between the 2 arms.



Fig 1. Screening, randomization, and follow-up of participants in a trial of azithromycin for infants aged 5–12 weeks of age.

https://doi.org/10.1371/journal.pmed.1004345.g001

| | Azithromycin (N = 16,416) | Placebo (<i>N</i> = 16,461) |
|-----------------------------|---------------------------|------------------------------|
| Age, days Mean (SD) | 48.9 (15.5) | 49.3 (15.4) |
| Sex, N (%) | | |
| Female | 8,045 (49%) | 8,136 (49%) |
| Male | 8,371 (51%) | 8,325 (51%) |
| District, N (%) | | |
| Nouna | 14,231 (87%) | 14,271 (87%) |
| Banfora | 1,298 (8%) | 1,305 (8%) |
| Karankasso-Vigue | 887 (5%) | 885 (5%) |
| Season of enrollment, N (%) | | |
| Rainy (June-October) | 7,174 (44%) | 7,279 (44%) |
| Dry (November–May) | 9,242 (56%) | 9,182 (56%) |
| Weight at enrollment, kg | | |
| Mean (SD) | 4.6 (0.9) | 4.6 (0.9) |
| Length at enrollment, cm | | |
| Mean (SD) | 55.3 (3.0) | 55.3 (3.1) |
| WLZ | | |
| Mean (SD) | -0.2 (1.5) | -0.2 (1.5) |
| WAZ | | |
| Mean (SD) | -0.6 (1.2) | -0.6 (1.2) |
| LAZ | | |
| Mean (SD) | -0.5 (1.3) | -0.5 (1.3) |
| MUAC | | |
| Median (IQR) | 12.1 (1.2) | 12.1 (1.2) |

| Table 1. | Baseline | characteristics | by | randomized | treatment | group. |
|----------|----------|-----------------|----|------------|-----------|--------|
|----------|----------|-----------------|----|------------|-----------|--------|

IQR, interquartile range; LAZ, length-for-age Z-score; MUAC, mid-upper arm circumference; SD, standard deviation; WAZ, weight-for-age Z-score; WLZ, weight-for-length Z-score.

https://doi.org/10.1371/journal.pmed.1004345.t001

Similarly, we found no evidence of a difference in the prevalence of underweight, wasting, or stunting in infants receiving azithromycin compared to placebo at 6 months of age (Table 3).

In a series of post hoc subgroup analyses, we found little consistent evidence of a difference in anthropometric endpoints in subgroups of children defined by age, sex, season, and baseline anthropometric status (**Figs 2** and **3** and **Tables B-G in S1 Appendix**). Subgroup analyses by age at enrollment overall found little evidence of a difference in growth endpoints by age, but infants enrolled in their third month of life who were randomized to azithromycin versus placebo had greater gains in length between enrollment and 6 months compared to those enrolled in the second month (mean difference for second month: -0.04 mm/day, 95% confidence interval (CI) -0.1 to 0.01; mean difference for second month: 0.1 mm/day, 95% CI 0.003 to 0.2, P for interaction = 0.008; **Table B in S1 Appendix**). Infants who were wasted at baseline (WLZ < -2) who received azithromycin had reduced odds of stunting at 6 months (OR 0.78, 95% CI 0.61 to 0.99, P for interaction = 0.04) compared to those who received placebo (**Table F in S1 Appendix**). Infants who were stunted at baseline who received azithromycin had reduced weight gain compared to those who received placebo (mean difference -0.6 g/day, 95% CI -1.1 to -0.04, P for interaction = 0.006; **Table G in S1 Appendix**).

Discussion

Overall, we found no evidence of a difference in growth outcomes at 6 months of age in infants receiving a single oral dose of azithromycin compared to placebo. These results accord with

| | Azithromycin Mean (SD) | Placebo Mean (SD) | Mean Difference (95% CI) | P value |
|-------------------------------------|------------------------|-------------------|--------------------------|---------|
| N | 13,641 | 13,657 | | |
| Age at follow-up, days ¹ | 185 (17) | 185 (17) | | |
| Weight gain, g/day | 18.0 (6.2) | 18.0 (6.4) | 0.03 (-0.12 to 0.18) | 0.67 |
| Height change, mm/day | 8.0 (2.1) | 8.0 (2.3) | 0.004 (-0.05 to 0.06) | 0.87 |
| Weight, kg ² | 7.1 (0.95) | 7.1 (0.97) | 0.005 (-0.01 to 0.02) | 0.63 |
| Length, cm ² | 66.1 (2.9) | 66.1 (2.9) | 0.01 (-0.05 to 0.08) | 0.66 |
| WAZ ² | -0.74 (1.1) | -0.74 (1.1) | -0.004 (-0.03 to 0.02) | 0.70 |
| WLZ ² | -0.64 (1.2) | -0.64 (1.2) | 0.001 (-0.03 to 0.03) | 0.94 |
| LAZ ² | -0.31 (1.3) | -0.31 (1.3) | -0.005 (-0.03 to 0.02) | 0.73 |
| MUAC, cm ² | 13.8 (1.2) | 13.8 (1.2) | 0.01 (-0.01 to 0.04) | 0.37 |

Table 2. Anthropometric outcomes by randomized treatment group.

CI, confidence interval; LAZ, length-for-age Z-score; MUAC, mid-upper arm circumference; SD, standard deviation; WAZ, weight-for-age Z-score; WLZ, weight-for-length Z-score.

¹Infants were considered measured "in window" if they were between 120 and 270 days of age at the 6-month follow-up visit.

²Adjusted for baseline value of the outcome.

https://doi.org/10.1371/journal.pmed.1004345.t002

previous results of azithromycin administration during the neonatal period that found no evidence of an effect of single-dose azithromycin on 6-month anthropometric measurements in Burkina Faso [18]. Similarly, studies of mass distribution of azithromycin for prevention of mortality and for trachoma control have generally not found evidence of a growth benefit from single-dose azithromycin among children under 5 years of age [19–22]. Taken together, these results suggest that in the general population, single-dose azithromycin administered during infancy does not improve growth outcomes in the months following administration.

The current trial collected anthropometric measurements at baseline (5 to 12 weeks of age) and at 6 months of age, limiting ability to make inferences about shorter-term or longer-term changes in anthropometric outcomes. However, infants were enrolled between 5 and 12 weeks of age, meaning that the 6-month endpoint ranged between 3 and 5 months after treatment for most infants. In subgroup analyses by age in months, there was little evidence of a difference in growth endpoints by age, although infants in the third month of life had greater height change with azithromycin compared to placebo versus those in the second month of life, although absolute differences were small. It is possible that azithromycin has some short-term effects that lead to short-term increases in height that do not persist or that there are benefits in treating older infants that are not observed in younger infants. However, multiple subgroup analyses were conducted, and thus, this finding may also be due to chance.

Randomized controlled trials of antibiotics for children with severe acute malnutrition have found a growth-promoting effect of antibiotics in the short term when antibiotics are included

| | Azithromycin N with outcome (%) | Placebo N with outcome (%) | Odds Ratio (95% CI) | P value |
|-------------------------|---------------------------------|----------------------------|---------------------|---------|
| N | 13,641 | 13,657 | | |
| Underweight (WAZ < -2) | 1,656 (12.1%) | 1,702 (12.5%) | 0.97 (0.90 to 1.04) | 0.42 |
| Wasted (WLZ < -2) | 1,701 (12.5%) | 1,714 (12.6%) | 0.99 (0.92 to 1.07) | 0.84 |
| Wasted (MUAC < 12.5 cm) | 1,222 (9.0%) | 1,265 (9.3%) | 0.96 (0.89 to 1.05) | 0.38 |
| Stunted (LAZ < -2) | 1,137 (8.3%) | 1,149 (8.4%) | 0.99 (0.91 to 1.08) | 0.82 |

Table 3. Wasting, underweight, and stunting by randomized treatment group.

CI, confidence interval; LAZ, length-for-age Z-score; MUAC, mid-upper arm circumference; WAZ, weight-for-age Z-score; WLZ, weight-for-length Z-score.

https://doi.org/10.1371/journal.pmed.1004345.t003



Fig 2. Differences in continuous anthropometric endpoints at 6 months of age in azithromycin compared to placebo among subgroups of infants defined by age at enrollment (grey bars), season of enrollment (orange bars), sex (blue bars), stunting (green bars), underweight (yellow bars), and wasting (black bars). Outcomes include weight gain in g/day (A), length change in mm/day (B), MUAC in cm (C), WAZ (D), LAZ (E), and WLZ (F). Means indicate mean value for the outcome in each subgroup by treatment arm, and N indicates the number of infants in each subgroup in each treatment arm. Mean differences for each outcome can be found in <u>S1 Appendix</u>. LAZ, length-for-age Z-score; MUAC, mid-upper arm circumference; WAZ, weight-for-age Z-score; WLZ, weight-for-length Z-score.

https://doi.org/10.1371/journal.pmed.1004345.g002

as part of a treatment program for uncomplicated severe acute malnutrition compared to amoxicillin [7,28]. However, these differences were typically only apparent in the weeks following treatment, with no evidence of a difference in weight gain several months after treatment administration [29]. Currently, most treatment guidelines for children with undernutrition are focused on children over 6 months of age. Thus, the role of antibiotics in





https://doi.org/10.1371/journal.pmed.1004345.g003

infants <6 months of age who are undernourished is unclear. In subgroup analyses, we were unable to demonstrate a difference in anthropometric outcomes following azithromycin administration in infants who were underweight (WAZ < -2) at baseline. We found contradictory results between infants with wasting and stunting at baseline, with infants with wasting randomized to azithromycin having reduced odds of stunting, but infants with stunting randomized to azithromycin gaining less weight. These results do not clearly indicate a benefit of early infancy antibiotics for children with undernutrition.

A major concern related to early life antibiotic administration has been the risk of weight gain and obesity in exposed children in high-income settings [3–5]. Antibiotics have been shown to be growth promoting in livestock [30,31]. This mechanism may be via treatment of infection or alteration of the gut microbiome. In the present analysis, while we did not show evidence of a difference in weight gain in infants receiving azithromycin compared to placebo, we are unable to comment on obesity, as the prevalence of overweight and obesity was too low in this population for meaningful inferences and infants were only followed until 6 months of age. Weight gain among infants in rural Burkina Faso is typically considered to be beneficial, and thus, any improvement in weight gain could be viewed as health promoting in this population. While the absence of increases in weight gain following azithromycin compared to placebo suggests it is unlikely that single-dose azithromycin causes obesity among infants, it is unclear if these results would generalize to populations with a higher prevalence of overweight among infants.

This study has several limitations. First, although the overall sample size was large, the sample size for some subgroups was limited. In particular, we did not have statistical power to

evaluate subgroups with severe underweight, wasting, or stunting (WAZ, WLZ, and LAZ < -3, respectively) or concurrent wasting and stunting. Infants in these categories have the highest risk of mortality, morbidity, and poor developmental outcomes and are a priority population for intervention. We are unable to comment on the role of single-dose azithromycin in these groups of children. Second, multiple subgroup analyses were conducted, and thus, it is possible that statistically significant findings were due to chance. Most differences were small and not clinically meaningful. The results of the subgroup analyses should be treated as hypothesis generating. Third, we collected anthropometric measurements only at enrollment (5 to 12 weeks of age) and 6 months of age. Any more proximal or long-term effects of azithromycin on anthropometric endpoints in infants would not be detected in these analyses. However, it is unlikely that a long-term difference in growth would be observed given the lack of effect at 6 months, and the clinical significance of any shorter-term changes is unclear given the lack of effect at 6 months. Fourth, this study evaluated a single 20 mg/kg dose of azithromycin compared to placebo, which is the dosing currently being considered for programmatic scale-up for prevention of child mortality and is the dosing used by trachoma programs for children [32,33]. Higher doses or repeated exposure to antibiotics may result in different outcomes, and we cannot comment on the effect of different antibiotics or different dosing regimens. Previous work in the study area has suggested that infants receive several courses of antibiotics per year [34,35]. Future studies considering the impact of repeated courses of antibiotics may yield additional insights. Finally, these results are likely only generalizable to populations of infants with a similar distribution of weight and height and with similar dietary and disease patterns.

Overall, we found no evidence of an effect of infant administration of a single dose of oral azithromycin compared to placebo on weight- or height-based outcomes. These results confirm earlier work evaluating single-dose azithromycin administered during the neonatal period [18]. Taken together, the 2 trials, which, in total, enrolled over 50,000 infants, provide definitive evidence that single-dose azithromycin does not cause weight or height changes in infants in Burkina Faso.

Supporting information

S1 Appendix. Supporting data tables. Table A. Baseline characteristics among infants included in the analysis (N = 27,743) and not included in the analysis (N = 5,134) by randomized treatment assignment. **Table B.** Results of subgroup analyses for each outcome at 6 months of age by age at enrollment in months. **Table C.** Results of subgroup analyses for each outcome at 6 months of age by child's sex. **Table D.** Results of subgroup analyses for each outcome at 6 months of age by season of enrollment. **Table E.** Results of subgroup analyses for each outcome at 6 months of age by underweight (weight-for-age Z-score < -2) at enrollment. **Table F.** Results of subgroup analyses for each outcome at 6 months of subgroup analyses for each outcome at 6 months of age by stunting (height-for-length Z-score < -2) at enrollment. **Table G.** Results of subgroup analyses for each outcome at 6 months of age by stunting (height-for-length Z-score < -2) at enrollment. **Table G.** Results of subgroup analyses for each outcome at 6 months of age by stunting (height-for-length Z-score < -2) at enrollment.

S2 Appendix. Statistical analysis plan for the trial. (PDF)

Acknowledgments

Azithromycin and matching placebo were donated by Pfizer (New York, NY).

Author Contributions

- **Conceptualization:** Ali Sié, Elodie Lebas, Kieran S. O'Brien, Thomas M. Lietman, Catherine E. Oldenburg.
- **Data curation:** Ali Sié, Mamadou Ouattara, Mamadou Bountogo, Clarisse Dah, Valentin Boudo, Huiyu Hu, Benjamin F. Arnold, Thomas M. Lietman, Catherine E. Oldenburg.
- Formal analysis: Valentin Boudo, Huiyu Hu, Benjamin F. Arnold, Catherine E. Oldenburg.
- Funding acquisition: Ali Sié, Mamadou Bountogo, Elodie Lebas, Thomas M. Lietman, Catherine E. Oldenburg.
- Investigation: Ali Sié, Mamadou Ouattara, Mamadou Bountogo, Clarisse Dah, Thierry Ouedraogo, Elodie Lebas, Benjamin F. Arnold, Kieran S. O'Brien, Thomas M. Lietman, Catherine E. Oldenburg.
- Methodology: Ali Sié, Mamadou Ouattara, Mamadou Bountogo, Clarisse Dah, Thierry Ouedraogo, Elodie Lebas, Benjamin F. Arnold, Kieran S. O'Brien, Thomas M. Lietman, Catherine E. Oldenburg.
- Project administration: Ali Sié, Mamadou Ouattara, Mamadou Bountogo, Clarisse Dah, Thierry Ouedraogo, Valentin Boudo, Elodie Lebas, Huiyu Hu, Benjamin F. Arnold, Kieran S. O'Brien, Thomas M. Lietman, Catherine E. Oldenburg.
- **Resources:** Ali Sié, Mamadou Ouattara, Mamadou Bountogo, Clarisse Dah, Thierry Ouedraogo, Valentin Boudo, Elodie Lebas, Huiyu Hu, Benjamin F. Arnold, Kieran S. O'Brien, Thomas M. Lietman, Catherine E. Oldenburg.
- Software: Valentin Boudo, Huiyu Hu, Benjamin F. Arnold.
- **Supervision:** Ali Sié, Mamadou Ouattara, Mamadou Bountogo, Clarisse Dah, Thierry Ouedraogo, Elodie Lebas, Huiyu Hu, Benjamin F. Arnold, Kieran S. O'Brien, Thomas M. Lietman, Catherine E. Oldenburg.
- Validation: Valentin Boudo, Huiyu Hu, Catherine E. Oldenburg.
- Visualization: Huiyu Hu, Catherine E. Oldenburg.
- Writing original draft: Ali Sié, Catherine E. Oldenburg.
- Writing review & editing: Ali Sié, Mamadou Ouattara, Mamadou Bountogo, Clarisse Dah, Thierry Ouedraogo, Valentin Boudo, Elodie Lebas, Huiyu Hu, Benjamin F. Arnold, Kieran S. O'Brien, Thomas M. Lietman, Catherine E. Oldenburg.

References

- Oldenburg CE, Sie A, Coulibaly B, Ouermi L, Dah C, Tapsoba C, et al. Effect of commonly-used pediatric antibiotics on gut microbial diversity in preschool children in Burkina Faso: A randomized clinical trial. Open Forum. Infect Dis Ther. 2018; 5(11):ofy289. https://doi.org/10.1093/ofid/ofy289 PMID: 30515431
- Cox LM. Antibiotics shape microbiota and weight gain across the animal kingdom. Anim Front. 2016; 6 (3):8–14. Available from: https://academic.oup.com/af/article/6/3/8/4638757.
- Scott FI, Horton DB, Mamtani R, Haynes K, Goldberg DS, Lee DY, et al. Administration of Antibiotics to Children Before Age 2 Years Increases Risk for Childhood Obesity. Gastroenterol Int. 2016 Jul 1; 151 (1):120–129. Available from: http://linkinghub.elsevier.com/retrieve/pii/S0016508516003528 https:// doi.org/10.1053/j.gastro.2016.03.006 PMID: 27003602
- 4. Tian J, Liu H, Guo H, Han W, Ding H, Chen T. Application of antibiotics before 3 years of age increases the risk of childhood overweight and obesity. Exp Ther Med. 2020; 21(1):1–7.

- Johnston BC, Srivastava A, Chau K, Kwon H, Guo Q. Early and frequent exposure to antibiotics in children and the risk of obesity: Systematic review and meta-analysis of observational studies. F1000Res. 2020:9.
- Gough EK, Moodie EEM, Prendergast AJ, Johnson SMA, Humphrey JH, Stoltzfus RJ, et al. The impact of antibiotics on growth in children in low and middle income countries: systematic review and metaanalysis of randomised controlled trials. BMJ. 2014 Apr 15; 348:g2267. Available from: http://www.bmj. com/cgi/doi/10.1136/bmj.g2267 PMID: 24735883
- Isanaka S, Langendorf C, Berthé F, Gnegne S, Li N, Ousmane N, et al. Routine Amoxicillin for Uncomplicated Severe Acute Malnutrition in Children. N Engl J Med. 2016 Feb 4; 374(5):444–453. Available from: http://www.nejm.org/doi/10.1056/NEJMoa1507024 PMID: 26840134
- Mussa A, Powis KM, Lockman S, Ajibola G, Morroni C, Smeaton L, et al. Prolonged Cotrimoxazole Prophylaxis Has No Impact on Child Growth in the First Two Years of Life: Findings from a Randomized Controlled Trial in Botswana. J Pediatr. 2022 Jul 1; 246:266–270.e2. https://doi.org/10.1016/j.jpeds. 2022.03.031 PMID: 35351531
- Toledo G, Landes M, van Lettow M, Tippett Barr BA, Bailey H, Crichton S, et al. Risk factors for stunting in children who are HIV-exposed and uninfected after Option B+ implementation in Malawi. Matern Child Nutr. 2023 Jan 1; 19(1). https://doi.org/10.1111/mcn.13451 PMID: 36349962
- Chelimo C, Camargo CA, Morton SMB, Grant CC. Association of Repeated Antibiotic Exposure up to Age 4 Years with Body Mass at Age 4.5 Years. JAMA Netw Open. 2020 Jan 22; 3(1). https://doi.org/10. 1001/jamanetworkopen.2019.17577 PMID: 31968112
- Block JP, Charles Bailey L, Gillman MW, Lunsford D, Daley MF, Eneli I, et al. Early Antibiotic Exposure and Weight Outcomes in Young Children. Pediatrics. 2018; 142(6):e20180290. <u>https://doi.org/10.1542/ peds.2018-0290</u> PMID: 30381474
- Black RE, Victora CG, Walker SP, Bhutta ZA, Christian P, de Onis M, et al. Maternal and child undernutrition and overweight in low-income and middle-income countries. Lancet. 2013 Aug 3; 382 (9890):427–451. https://doi.org/10.1016/S0140-6736(13)60937-X PMID: 23746772
- O'Brien KS, Amza A, Kadri B, Nassirou B, Cotter SY, Stoller NE, et al. Comparison of anthropometric indicators to predict mortality in a population-based study of children under 5 in Niger. Public Health Nutr. 2020; 23 (3):538–543.
- Hoehn C, Lelijveld N, Mwangome M, Berkley JA, McGrath M, Kerac M. Anthropometric Criteria for Identifying Infants Under 6 Months of Age at Risk of Morbidity and Mortality: A Systematic Review. Clin Med Insights Pediatr. 2021 Jan; 15:117955652110499.
- Mertens A, Benjamin-Chung J, Colford JM, Coyle J, van der Laan MJ, Hubbard AE, et al. Causes and consequences of child growth faltering in low-resource settings. Nature. 2023 Sep 21; 621:568–576. https://doi.org/10.1038/s41586-023-06501-x PMID: 37704722
- Keenan JD, Bailey RL, West SK, Arzika A, Hart J, Weaver J, et al. Azithromycin to Reduce Childhood Mortality in Sub-Saharan Africa. N Engl J Med. 2018 Feb 28; 378(17):1583–1592. <u>https://doi.org/10.1056/NEJMoa1715474</u> PMID: 29694816
- Oldenburg CE, Sie A, Bountogo M, Zakane A, Compaore G, Ouedraogo T, et al. Neonatal azithromycin administration for prevention of infant mortality. NEJM Evid. 2022; 1(4):EVIDoa2100054. <u>https://doi.org/10.1056/EVIDoa2100054</u> PMID: 35692260
- Sie A, Bountogo M, Zakane A, Compaoré G, Ouedraogo T, Ouattara M, et al. Neonatal Azithromycin Administration and Growth during Infancy: A Randomized Controlled Trial. Am J Trop Med Hyg. 2023 May 1; 108(5):1063–1070. https://doi.org/10.4269/ajtmh.22-0763 PMID: 36972694
- Arzika AM, Maliki R, Ali MM, Alio MK, Abdou A, Cotter SY, et al. Effect of Mass Azithromycin Distributions on Childhood Growth in Niger: A Cluster-Randomized Trial. JAMA Netw Open. 2021 Dec 30; 4 (12):e2139351. https://doi.org/10.1001/jamanetworkopen.2021.39351 PMID: 34967883
- Amza A, Yu SN, Kadri B, Nassirou B, Stoller NE, Zhou Z, et al. Does Mass Azithromycin Distribution Impact Child Growth and Nutrition in Niger? A Cluster-Randomized Trial. King JD, editor. PLoS Negl Trop Dis. 2014 Sep 11; 8(9):e3128. https://doi.org/10.1371/journal.pntd.0003128 PMID: 25210836
- Keenan JD, Gebresillasie S, Stoller NE, Haile BA, Tadesse Z, Cotter SY, et al. Linear growth in preschool children treated with mass azithromycin distributions for trachoma: A cluster-randomized trial. PLoS Negl Trop Dis. 2019; 13(6):e0007442. <u>https://doi.org/10.1371/journal.pntd.0007442</u> PMID: 31166952
- 22. Amza A, Kadri B, Nassirou B, Stoller NE, Yu SN, Zhou Z, et al. A cluster-randomized controlled trial evaluating the effects of mass azithromycin treatment on growth and nutrition in Niger. Am J Trop Med Hyg. 2013 Jan 1; 88(1):138–143. Available from: http://www.ajtmh.org/cgi/doi/10.4269/ajtmh.2012.12-0284 PMID: 23208876
- 23. Bountogo M, Sié A, Zakane A, Compaoré G, Ouédraogo T, Brogdon J, et al. Infant mortality and growth failure after oral azithromycin among low birthweight and underweight neonates: A subgroup analysis of

a randomized controlled trial. Robinson J, editor. PLoS Glob Public Health. 2023 May 15; 3(5): e0001009. https://doi.org/10.1371/journal.pgph.0001009 PMID: 37186577

- Sié A, Ouattara M, Bountogo M, Bagagnan C, Coulibaly B, Boudo V, et al. A double-masked placebocontrolled trial of azithromycin to prevent child mortality in Burkina Faso, West Africa: Community Health with Azithromycin Trial (CHAT) study protocol. Trials. 2019; 20:675. <u>https://doi.org/10.1186/</u> s13063-019-3855-9 PMID: 31801563
- Sié A, Ouattara M, Bountogo M, Boudo V, Ouedraogo T, Compaoré G, et al. Azithromycin during Routine Well-Infant Visits to Prevent Death. N Engl J Med. 2024; 390(3):221–229. <u>https://doi.org/10.1056/</u> NEJMoa2309495 PMID: 38231623
- 26. Ouattara M, Sié A, Bountogo M, Boudo V, Dah C, Lebas E, et al. Anthropometric Differences in Community- Versus Clinic-Recruited Infants Participating in a Trial of Azithromycin for Prevention of Childhood Mortality in Burkina Faso. Am J Trop Med Hyg. 2023 Oct 2. Available from: https://www.ajtmh.org/view/journals/tpmd/aop/article-10.4269-ajtmh.23-0298/article-10.4269-ajtmh.23-0298.xml
- 27. WHO Multicentre Growth Reference Study Group. WHO Child Growth Standards: Length/height-forage, weight-for-age, weight-for-length, weight-for-height and body mass index-for-age: Methods and development. Geneva: World Health Organization. 2006;312.
- Trehan I, Goldbach HS, LaGrone LN, Meuli GJ, Wang RJ, Maleta KM, et al. Antibiotics as Part of the Management of Severe Acute Malnutrition. N Engl J Med 2013 Jan 31; 368(5):425–435. Available from: http://www.nejm.org/doi/10.1056/NEJMoa1202851 PMID: 23363496
- Isanaka S, Grantz KH, Berthe F, Schaefer M, Adehossi E, Grais RF. Extended Follow-up From a Randomized Clinical Trial of Routine Amoxicillin in the Treatment of Uncomplicated Severe Acute Malnutrition in Niger. JAMA Pediatr. 2020; 174 (3):295–297. https://doi.org/10.1001/jamapediatrics.2019.5189 PMID: 31930369
- Agunos A, Léger DF, Carson CA, Gow SP, Bosman A, Irwin RJ, et al. Antimicrobial use surveillance in broiler chicken flocks in Canada, 2013–2015. PLoS ONE. 2017; 12(6):e0179384. <u>https://doi.org/10. 1371/journal.pone.0179384</u> PMID: 28658278
- **31.** Cromwell GL. Why and how antibiotics are used in swine production. Anim Biotechnol. 2002 May; 13 (1):7–27. https://doi.org/10.1081/ABIO-120005767 PMID: 12212945
- 32. Solomon AW, Burton MJ, Gower EW, Harding-Esch EM, Oldenburg CE, Taylor HR, et al. Trachoma. Nat Rev Dis Primers. 2022 Dec 26; 8(1):32. Available from: https://www.nature.com/articles/s41572-022-00359-5 https://doi.org/10.1038/s41572-022-00359-5 PMID: 35618795
- World Health Organization. WHO Guideline on mass drug administration of azithromycin to children under five years of age to promote child survival [Internet]. Geneva; 2020 [cited 2022 Sep 14]. Available from: https://www.who.int/publications/i/item/9789240009585
- Sie A, Coulibaly B, Adama S, Ouermi L, Dah C, Tapsoba C, et al. Antibiotic prescriptions among children under age 5 in Nouna District, Burkina Faso. Am J Trop Med Hyg. 2019; 100 (5):1121–1124.
- Sié A, Ouattara M, Bountogo M, Dah C, Compaoré G, Boudo V, et al. Indication for Antibiotic Prescription tion Among Children Attending Primary Healthcare Services in Rural Burkina Faso. Clin Infect Dis. 2021; 73(7):1288–1291. https://doi.org/10.1093/cid/ciab471 PMID: 34018004